Fri Apr 19 16:48:39 GMIr 2002
Year: 02 Doy: 109
Qobserver: yasukawa
WEATHER COMMVENT: Fri Apr 19 16:48:40 GJr 2002

cool, clear, light southwest breeze.
Fri Apr 19 16:50:17 GJIr 2002 CH P Start Patrol
Fri Apr 19 16:50:37 GMVI 2002 Pl CS Start Patrol

Fri Apr 19 16:50:56 GMI 2002 MKI V Start Patro

**P| CS PROBLEM *: Fri Apr 19 16:57:49 GMI 2002

PI CS obj ective wheel still flaky.

Fri Apr 19 17:07:18 GMVIr 2002 Pl CS End Patr ol

PSPT COMMENT: Fri Apr 19 17:26:47 GMI 2002

Starting observati ons.

**ECHO PROBLEM *: Fri Apr 19 17:27:17 GMI 2002

| mmge down in |ower right corner.

**ECHO PROBLEM *: Fri Apr 19 17:28:59 GMI 2002

Restarting echosys. HA error was -0.001 and DEC error was -0.000

**P| CS PROBLEM *: Fri Apr 19 17:47:55 GMI 2002

Checked out status board again and checked out hone switch. Both appear OK
It looks like the failed conponent m ght be the HEl optical encoder.
Swapping it out.

Fri Apr 19 18:35:37 GVI 2002 Pl CS Start Patrol

**P| CS PROBLEM*: Fri Apr 19 18:35:54 GV 2002

HElI optical encoder replaced with a new unit. Tests w th handheld controller
i ndicate problemis fixed.

**ECHO PROBLEM *: Fri Apr 19 18:36:57 GMI' 2002

Image is still offset in window. Restarting again.

A NEW TAPE HAS BEEN PUT | NTO KAI EE DLT DRIVE, Fri Apr 19 18:45:44 GMI 2002
Fri Apr 19 19:01:02 GMVr 2002 CH P LSD

Fri Apr 19 19:03:18 GMVIr 2002 CH P End LSD

Fri Apr 19 19:03:32 GV 2002 CH P Bi asLSD

Fri Apr 19 19:04:35 GVI 2002 CH P End Bi asLSD

Fri Apr 19 19:04:49 GJI 2002 CH P Bi as

Fri Apr 19 19:05:48 GVIr 2002 CH P End Bi as

WEATHER COMMENT: Fri Apr 19 19:10:49 GMI 2002

I n orographic clouds.

PSPT COMMENT: Fri Apr 19 19:12: 03 GMI 2002

St opped observati ons.

WEATHER COMMENT: Fri Apr 19 19:22:23 GMI' 2002

Solid overcast. Cosing done in case of precipitation.
Fri Apr 19 19:23:13 GMJr 2002 CH P End Patr ol
**MKIV PROBLEMF*: Fri Apr 19 19:23:09 GMI 2002

Did not calibrate nk4 due to flurry of other activity so far this norning.
Fri Apr 19 19:24:21 GJIr 2002 Pl CS End Patr ol

Fri Apr 19 19:24:40 GV 2002 MKI V End Patr ol



MKI'V COMMENT: Fri Apr 19 21:35:24 GV 2002
Connected the RTD and heater strip to the controller for FLC heater. Secured
cabl es inside the troughs. Secured the cable run between the wap-up
and the AC swi tchbox via the service | oop. Added the power cable to switch
box. Reinstalled the printing tenperature nonitor and its therm stor.
Buttoned up nk4. Waiting for 72-hour wet-out period to end before
firing up the new FLC heater system should be Saturday 4/20.
WEATHER COMMENT: Fri Apr 19 21:43:16 GMI 2002
Still in orographic cloud overcast and Iight fog.
**ECHO PROBLEMF*: Fri Apr 19 21:43:59 GMIr 2002
Did not get a chance to see if image position got fixed by restart, clouds
too thick.
Fri Apr 19 21:46:37 GVI 2002

IV

16 50.rawk4 17 23.rawnrk4 17 55.rawrk4 18 28.rawnk4 19 00. rawnk4
16 53.rawk4 17 26.rawrk4 17 58.rawnrk4 18 31.rawnk4 19 03.rawrk4
16 _56.rawrk4 17 _29.rawnk4 18 01.rawrk4 18 34.rawrkd4d 19 06.rawnk4
16 _59.rawrk4 17 32.rawnk4 18 04.rawrk4 18 37.rawrkd4d 19 09.rawnk4
17 _02.rawrk4 17 35.rawnk4 18 07.rawrk4 18 40.rawrk4d 19 12.rawnk4
17_05.rawrk4 17 _38.rawnk4 18 10.rawrk4 18 43.rawrk4d 19 15.rawnrk4
17 08.rawk4 17 41.rawrk4 18 13.rawrkd4 18 46.rawnk4 19 18.rawnk4
17 11.rawk4 17 44.rawrk4 18 16.rawrk4 18 48.rawnk4d 19 21.rawk4
17 _14.rawrk4 17 _47.rawk4 18 19.rawnk4 18 51.rawnk4

17 _17.rawrk4 17 _50.rawnk4 18 22.rawnk4 18 54.rawnk4

17_20.rawrk4 17_52.rawrk4 18 _25.rawrk4 18_57.rawrk4



